

Temi di Discussione

(Working Papers)

Inward foreign direct investment and innovation: evidence from Italian provinces

by Roberto Antonietti, Raffaello Bronzini and Giulio Cainelli







Temi di discussione

(Working papers)

Inward foreign direct investment and innovation: evidence from Italian provinces

by Roberto Antonietti, Raffaello Bronzini and Giulio Cainelli

Number 1006 - March 2015

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: GIUSEPPE FERRERO, PIETRO TOMMASINO, PIERGIORGIO ALESSANDRI, MARGHERITA BOTTERO, LORENZO BURLON, GIUSEPPE CAPPELLETTI, STEFANO FEDERICO, FRANCESCO MANARESI, ELISABETTA OLIVIERI, ROBERTO PIAZZA, MARTINO TASSO. *Editorial Assistants:* ROBERTO MARANO, NICOLETTA OLIVANTI.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

INWARD FOREIGN DIRECT INVESTMENT AND INNOVATION: EVIDENCE FROM ITALIAN PROVINCES

by Roberto Antonietti⁶, Raffaello Bronzini^{*} and Giulio Cainelli^{*}

Abstract

This paper investigates empirically whether inward greenfield foreign direct investment (FDI) is related to greater sectorial innovation in the host Italian provinces. We combine several sources of data to estimate panel count models, regressing the annual number of patents in each province and industry against a series of lagged FDI variables. Our results show that a positive relationship between FDI and local patenting emerges only for services. In particular, we find that greater inward FDI in services positively influences local patenting activity in knowledge-intensive business services. These results are robust to endogeneity and the inclusion of province controls and fixed effects.

JEL Classification: F14, F23, O31, C23.

Keywords: inward greenfield FDI, innovation, patents, research and development, panel count models.

Contents

1. Introduction	5
2. Background literature	7
3. Empirical model and data	
3.1 Empirical model	
3.2 Data and variables	
4. Results	
5. Robustness checks	
6. Conclusions	
References	
Appendix	
11	

 ⁶ University of Padova, Department of Economics and Management; email: <u>roberto.antonietti@unipd.it</u>
 * Bank of Italy, Directorate General for Economics, Statistics and Research; email:

raffaello.bronzini@bancaditalia.it

^{*}University of Padova, Department of Economics and Management and CERIS-CNR; email: <u>giulio.cainelli@unipd.it</u>

1. Introduction¹

The effect of inward foreign direct investment (IFDI) on host countries has been the subject of a longstanding empirical investigation in the economics literature. The theory suggests two main effects of the entry of foreign firms on the host economy. First, FDI are considered an important channel of technology transfer because Multinational Enterprises (MNEs) are more productive, more innovative and invest more in research and development (R&D) than domestic enterprises. The entry of MNEs is supposed to benefit incumbent firms thanks to knowledge transmission fostered by vertical and horizontal linkages or knowledge spillovers (Blomström and Kokko 1998). However, technology transfer cannot be taken for granted. Rather, it will depend on the capacity of the incumbent firms to absorb and apply new technologies and, more generally, on the economic environment to allow knowledge transmission among foreign and domestic firms.

The economic theory suggests also that inward FDI might have a pro-competitive effect on the host economy. The entry of foreign firms boosts competition in the local market, pushing incumbent firms to search for productivity improvements and promoting the reallocation of resources toward more productive units (Keller 2009; Kiriyama 2012). However, if the resources released from domestic firms pushed out of the market are not quickly re-employed in more efficient firms, in the short term FDI might also have negative effects.

In many countries, to take advantage of the supposed gains from hosting MNEs, governments compete to attract foreign firms by offering tax reliefs, financial assistance and many other types of benefits. Since these policies imply large costs for public finance, they are justified only if the positive externalities on the host economy stemming from FDI are substantial. However, from an empirical point of view, the effect of FDI on host economies remains an open question (see Section 2 for a review).

While the impact of inward FDI on productivity has been investigated extensively in the empirical literature, including at regional level (see, among others.: Haskel *et al.* 2007; Peri and Urban 2006; Greenstone *et al.* 2010; Castellani and Pieri 2015), the effect on domestic innovation capabilities has received much less attention and, to our knowledge, there are no investigations of this effect at the local level.² The present paper fills this gap. We study the impact of Inward Greenfield FDI (IGFDI) on the inventive capability of narrow Italian territorial areas (the

¹ The views expressed in this paper are those of the authors and do not necessarily reflect those of their home institutes. We would like to thank for their helpful comments and useful suggestions: Davide Castellani, Maria Luisa Mancusi Lucia Piscitello, three anonymous referees and the participants at the conferences of Bank of Italy, University of Bari, Bocconi University and Utrecht University. We are very grateful to Francesca Lotti and Giovanni Marin for providing us with firm-level data on patent applications. The usual disclaimer applies.

² See Bertschek (1995), Branstetter (2006), Aghion et al. (2009), Brambilla et al. (2009) and Vahter (2011).

provinces-NUTS3), measured by their patenting activity. Theoretically, host country innovativeness can be affected by the entry of foreign firms through the mechanisms described above, namely, technology transfers and pro-competitive effects.

There are two modes of entry of a FDI in a region: cross-border Mergers and Acquisitions (M&A) and greenfield FDI. By M&A a foreign company takes over, or merges with, an existing domestic firm, whereas Greenfield FDI is defined as a new entry or an expansion in the host country of a foreign firm or plant. Thus, unlike cross-border M&A which involve existing domestic assets, greenfield FDI are investment projects that entail the establishment of new assets and activities in the host country (e.g. see: UNCTAD 2009; chapter III). Since greenfield FDI have an additional effect on the host-economy, it is more likely that they cause pro-competitive effects and technological spillovers within the host-country than cross-border M&A. For instance, by opening new productive plants in the domestic countries, or by expanding existing ones, a foreign enterprise will compete more strongly with domestic firms and might give a boost to their productivity improvements. For these reasons, we focus only on inward greenfield foreign direct investment because in our view they are more able to trigger these effects than cross-border M&A (here not considered). At the same time, our data take account of the size of the FDI in terms of capital flows or employment created, which is useful for empirical analysis to evaluate the impact of FDI based on the intensity of investment flows.³ As regards innovative capability, we use a hard measure of innovation given by the number of patent applications from incumbent firms to the European Patent Office (EPO).

To study the impact of FDI at the local level over the Italian territory is important for several reasons. First, technological transfer might have a local dimension. Knowledge transmission can be fostered by geographical proximity between MNEs and domestic firms (Jaffe *et al.* 1993) based, for example, on informal contacts, collaboration among firms, workers' mobility, and supply and demand linkages. All these channels are likely to be more effective among firms in the same location. Furthermore, it is likely also that pro-competitive effects on incumbent firms will be stronger if the foreign firms compete in the same local domestic market. Notice that, unlike most previous papers that examine the effects of FDI (on domestic productivity) from a regional perspective, we use a narrower geographic unit, the provinces, which correspond to the NUTS-3 Eurostat classification, more suitable to capture agglomeration economies and local spillovers.

³ Aghion et al. (2009) use a similar measure of FDI, namely the employment related to the annual greenfield foreign firm entry. With respect to Aghion et al. (2009) our measure includes also the expansions of existing foreign firms or plants, not only the entry of new foreign firms. In our view this is not a limitation of our data since also an expansion of existing foreign firms might cause pro-competitive effects on the incumbent enterprises as well as technological spillovers.

Second, innovation is considered a key driver of economic growth, but Italy lags behind most advanced countries in terms of innovation performance (European Commission 2013). In addition, among the main European Union countries, Italy is the most affected by geographical economic disparities. Therefore, understanding whether the entry of foreign enterprises improves the production of innovations in incumbent firms is particularly important to better deal with these regional disparities.

Using panel count data models and controlling for province, industry-specific fixed effects, and endogeneity with instrumental variable (IV) estimates, we find that inward FDI in service sector increases the numbers of local patents related to service activities, namely those produced by knowledge intensive business service (KIBS) firms. This result holds if FDI is measured through a simple dummy variable and also if we account for the intensity of the investment in terms of jobs created or capital investments. However, we do not find a significant relationship between innovation and IGFDI in the manufacturing activities.

The remainder of the paper is organized as follows. Section 2 presents an overview of the related literature. Section 3 describes the empirical model, the econometric strategy and the data employed. Section 4 discusses our baseline empirical results and Section 5 presents some robustness checks. Section 6 draws the main conclusions of our investigation.

2. Background literature

Our paper bridges the gap between two streams of literature, of which the most recent papers are summarized in Table 1. There is a strand of work on whether IFDI boosts the productivity of the incumbent firms in the same territorial area.⁴ These papers are grounded on theories of agglomeration economies, which argue that positive knowledge spillovers from FDI are more likely if the domestic and foreign firms are located in the same area since geographical proximity encourages the diffusion of ideas and technology, owing to personal contacts and/or transfer of workers across firms (Greenstone *et al.* 2010). In these papers the main empirical issue is related to the potential reverse causation between FDI and productivity: MNEs might invest more in the most productive regions because they are likely to provide the highest profits. In these circumstances, spurious correlations between inward FDI and host-region productivity could arise.

⁴ This is a branch of a wider literature on the impact of inward FDI on host firm productivity or economic growth which includes, among others, Haddad and Harrison (1993) and Aitken and Harrison (1999) who found a negative effect of FDI on domestic productivity, Javorcik (2004), Keller and Yeaple (2009), Bitzer and Görg (2009) who show a positive impact of FDI on host country productivity. Cipollina *et al.* (2012) find a positive effect on the growth of recipient sectors. See also Blomström and Kokko (1998) for a review.

The paper by Peri and Urban (2006) analyses the impact of IFDI on domestic firms' productivity in Italian and German regions. They show that the average level of productivity among the foreign-owned firms affects the productivity of incumbents located in the same region. In their view, what matters is the productivity gap between foreign and domestic firms, and their results demonstrate the scope for productive catch-up by domestic firms. They tackle the issue of endogeneity using lagged explanatory variables as instruments, according to standard dynamic panel estimation methods.

Haskel *et al.* (2007) test the impact of inward FDI (measured as share of foreign-owned plant employment in total employment by sector or region) on the productivity growth of domestic firms in the UK regions. They find that FDI spurs the productivity of incumbent firms in the same industry as the investment, but do not find a significant correlation between FDI and productivity growth among firms in the same region as the investment. They address the endogeneity issue using lagged measures for FDI, or the FDI directed to the US.⁵ In a more recent study, Greenstone *et al.* (2010) analyse the impact of large foreign plant entry in US counties on the productivity of incumbent firms. They show that positive spillovers due to agglomeration economies occur among firms that share the same type of workers or the same technology. However, their findings do not support the hypothesis that input-output linkages encourage productivity spillovers. Greenstone and colleagues estimate the model for recipient counties that host foreign plants and those that did not receive the foreign plants because were barely dismissed as hosting county by the foreign enterprise. Since the latter are very similar to the former in terms of observable and, probably, unobservable variables, it is assumed that the estimates are not affected by endogeneity bias.⁶

The second strand of work investigates the impact of IFDI on incumbent firms' innovation capabilities. Most of these papers focus on the mechanism of increased competition due to the entry of MNEs into the domestic market. Stronger competition can encourage incumbent firms to improve their competitiveness by increased innovation and spur the reallocation of resources toward more competitive and more innovative firms. In the case of German firms, Bertschek (1995) and Blind and Jungmittag (2004) show that IFDI has a positive impact on product and process innovations in manufacturing and service firms, respectively. Aghion *et al.* (2009) find a positive effect of MNE entry on the number of UK domestic firms' patents in technologically advanced, but not traditional sectors. Aghion and colleagues apply an instrumental variables approach using a series of policy reforms to instrument the entry of foreign firms. The results provided in Brambilla

⁵ Similarly, for Venezuela, Aitken and Harrison (1999) found no evidence of productivity spillovers from FDI on domestic firms located in the same region as the MNEs.

⁶ Work on the effect of FDI on regional productivity includes among others: Driffield (2004) and Girma and Wakelin (2007) for UK regions, Bode and Nunnenkamp (2011) for the US and Castellani and Pieri (2015) for EU regions. See Table 1 for an overview.

et al. (2009) mostly support their findings. They observe that the probability of domestic Chinese firms introducing a product innovation increases with the presence of foreign firms in the same industry; moreover, since this overall effect is driven by less "sophisticated" firms (i.e. non-exporters, small employers, and small investors in R&D) they conclude that FDI mainly encourages imitation. Finally, Vahter (2011) examines the effect of FDI on innovation among domestic manufacturing firms in Estonia and finds that an increase in the share of FDI in one sector increases the probability that domestic firms in the same sector will introduce product or process innovations. Like Haskel *et al.* (2007), Vahter instruments inward FDI with FDI in the same industries in other Central and Eastern European economies.

Table 1. Recent studies of the effects of inward FDI on incumbent firms' productivity and innovation

Authors	Data	Unit of analysis	FDI measures	Econometric method	Results				
	Panel A. Dependent Variable: Productivity - Regional perspective								
Driffield (2004)	Industry- region data	UK regions	Stock of inward FDI	Dynamic panel data with lagged FDI	Positive effects only in non-assisted regions				
Peri and Urban (2006)	Aida and Reprint firm level data	Italian provinces, German regions	FDI concentration, productivity of foreign owned firms (same region or sector)	Dynamic panel data with lagged FDI	Positive effects only for productivity of foreign firms				
Girma and Wakelin (2007)	Plants of electronics industry	UK regions	Share of foreign firms (same region or sector)	Panel data with IV for FDI	Positive effects of FDI only in the same region				
Haskel <i>et al.</i> (2007)	Census	UK regions	Share of foreign firms (same region-sector)	Panel data with lagged FDI and IV	Positive effects of FDI only in the same industry				
Greenstone <i>et al.</i> (2010)	Census (TFP) and plant level data (FDI)	US Counties	Foreign firm entry	Panel data with control group	Positive effect				
Bode and Nunnenkamp (2011)	US BEA	US States	Density of FDI stock	Sigma convergence	Negative effect on convergence				
Castellani and Pieri (2015)	EU regional data set, fDI markets	EU regions	Greenfield inward and outward FDI	Panel data with lagged FDI	Positive effects				
		Panel B. Depend	dent Variable: Product or p	process innovation, p	atents				
Bertschek (1995)	Ifo, German Statistical Yearbook	German manufacturing firms	Share of inward FDI (same sector)	Panel Probit with lagged FDI	Positive effects				
Blind and Jungmittag (2004)	Innovation survey	German service firms	Lagged inward FDI stock	Probit	Positive effects				
Aghion <i>et al.</i> (2009)	UK national statistics	UK plants	Greenfield foreign firm entry (same sector)	Poisson and IV	Positive effects only in technologically advanced industries				
Brambilla <i>et al.</i> (2009)	World Bank data	Chinese firms	Share of foreign firms (same sector)	Linear probability model with f.e.	Positive effects				
Vahter (2011)	CIS, national Business Register	Estonian firms	Foreign firm entry (same sector)	Probit and IV	Positive effects				

Although several studies investigate the effect of foreign investment on the innovation propensity of domestic firms, ours is the only paper that studies this effect at the local level. Local level effects are interesting for a number of reasons. First, theoretical work on agglomeration economies highlights how geographic proximity can encourage the transmission of ideas owing, for example, to informal contacts or transfer of workers across firms. These information spillovers are best captured by studying domestic and foreign firms that share a local market. In the present paper, our territorial unit is the province (corresponding to the NUTS 3 level of European classification of territorial units), a smaller geographic unit than generally studied in the empirical literature which tends to focus on regions (NUTS 2). In our view, a province level study is better suited for capturing local spillovers. Second, the entry of a foreign firm can trigger pro-competitive effects on incumbent firms which are better studied using a narrow local lens, since the local market in the area targeted by FDI is likely to be the most affected by the entry of foreign enterprises. Finally, to the extent that the (hopefully positive) effects of entry of foreign firms are local in scope, IFDI could be important for understanding the innovation gap affecting many Italian regions and could also constitute a potential regional policy channel to reduce territorial disparities.

3. Empirical model and data

3.1 Empirical model

The empirical model corresponds to an equation that can be considered an extension of the knowledge production function (Griliches 1979), where provincial innovation is regressed on a series of innovation inputs including FDI. The model has the following implicit form:

[1]
$$Y_{S,P,T} = f(FDI_{P,T-1}, X_{S,P,T-1})$$

where Y_{SPT} is a measure of the innovation performance of industry *S* located in province *P* at time *T*; *FDI* is the inward greenfield FDI in province *P* at time *T*-1,⁷ and *X* is a vector of the additional time-lagged covariates.

Our measure of innovation is the number of patent applications submitted to the European Patent Office (EPO). There are advantages and disadvantages to measuring innovation by patents. On the one hand, it is well known that not all innovation are patented and there are informal mechanisms, such as secrecy or lead time advantages, that firms may use to appropriate returns

⁷ We use one-year lagged explanatory variables to mitigate (potential) simultaneity bias between patenting and FDI at the local level.

from their inventions. Also, the propensity to patent might differ across sectors and time. On the other hand, the patent is a hard measure of innovation. Compared to proxies based on survey data, such as number of new products and process introduced by the firms, patent applications suggest a higher quality of the innovation, because firms know that for being granted a patent requires accurate examination of the invention by experts who judge its novelty. Moreover, they are less prone to, even though not completely free from, personal or subjective considerations.⁸ In conclusion, we believe patent propensity to be a sound and powerful measure of innovation output suitable to evaluate the impact of FDI on host-country innovation capability, which has been also extensively used by the empirical literature on FDI and innovation (see e.g. Aghion et al. 2009).

Our empirical analysis uses several greenfield FDI variables. The first is a dummy (*DFDI*) that equals 1 when at least one FDI project is registered in a province P and year T-1, and 0 otherwise. The second is a continuous variable measuring the number of jobs created by foreign project investments in province P and year T-1 (*JOBS_FDI*). The third is a continuous variable measuring total greenfield foreign capital investment in the province P and the year T-1 (*KINV_FDI*_{p,t-1}). This provides information on the localization of FDI and also on its intensity. We split these last two variables according to the industry of the FDI investment: we distinguish between jobs created by FDI in the manufacturing industry of province P (*JOB_FDI_MAN*) and jobs created by FDI in the service industry of province P (*JOB_FDI_SER*). In addition, we distinguish between capital investment in manufacturing (*KINV_FDI_MAN*) and capital investment in services in province P (*KINV_FDI_SER*).⁹

In addition to FDI, we include a large set of controls in our estimation of equation 1. We include some determinants of innovation output such as the T-1 expenditure on R&D in province P and industry S (R&D) and (log) level of employment in province P, and sector S (EMPL) at time T-1. The latter variable is used as scaling factor to capture the existence of scale economies in the knowledge production function. Following the literature on the spatial determinants of innovation (Feldman and Audretsch 1999; Carlino *et al.* 2007; Knudsen *et al.* 2008; Beaudry and Schiffauerova 2009 for reviews) we include variables capturing urbanization and specialization economies. The former are proxied by the provincial population density (DEN) and a dummy that is equal to 1 for provinces that include large metropolitan areas or large urban zones (LUZ). The latter are measured by an index of provincial manufacturing specialization (SPEC MAN), given by the

 $^{^{8}}$ For example, the Community Innovation Survey – CIS, the main European survey of firm innovation, which is administered by Eurostat, considers firms to be innovative if they have produced a good that is new to the firm, but not necessarily to the market. Using patents as a measure ensures that we are capturing both firm level and market level innovation.

⁹ Since we observe many zero-values for FDI at province-sector level, we aggregate FDI for the whole manufacturing and service sectors (either jobs created or capital investment) at province level.

(log) share of manufacturing value-added in province *P* over total provincial value-added. Finally, we include a set of NUTS-1 regional dummies, industry dummies (Low Tech, Medium-Low Tech, Medium-High Tech, High-Tech and KIBS according to the OECD classification - see Appendix 1), and year dummies to capture region-specific, industry-specific and macroeconomic fixed effects.

The econometric strategy. Since we measure local innovation performance as y_{spT} , that is the number of patent applications from firms in sector *S* and province *P* in year *T*, we estimate a panel data count model. Following Baltagi (2005), the Poisson panel regression model is specified as follows:

[2]
$$\Pr(Y_{spT} = y_{spT} \mid x_{spT}) = \frac{e^{-\lambda_{spT}} \lambda_{spT}^{y_{spT}}}{y!^{spT}}$$

where y_{spT} (s=1, 2, ..., N; p=1, 2, ..., 103, T=2003, ..., 2008) denotes the number of occurrences of the event and λ is the mean and the variance of the distribution. Since we assume that y is the number of patent applications, λ will depend on a set of covariates that affect innovation propensity at the provincial industry level. We specify λ as a log-linear model such as $\ln \lambda_{spT} = \mu_{sp} + x'_{spT}\beta$, where μ_{sp} denotes unobservable individual specific effects.

The Poisson specification assumes equality between the mean and variance of the distribution, or the equidispersion property. If this hypothesis is rejected – the case generally defined as "overdispersion" – the Poisson specification is not appropriate and a negative binomial model specification is preferred. The panel data version was developed by Hausman *et al.* (1984). The negative binomial distribution has mean λ and variance $\lambda + \alpha \lambda^{2^{-k}}$. If $\alpha = 0$, a Poisson specification can be estimated. Therefore, we estimate a random-effect negative binomial model if α is significantly different from 0.¹⁰

3.2 Data and variables

The data for the empirical analysis come from several sources. We first match Italian patent applicants and Italian firms included in the AIDA database collected by Bureau van Dijk (Marin 2012). Original patent data come from the EPO Worldwide Patent Statistical database (PATSTAT) constructed by the EPO on behalf of the OECD Taskforce on Patent Statistics. This database

¹⁰ Note that we do not estimate a fixed-effects negative binomial model: as Guimaraes (2008) points out, this specification is able to control for 'true' fixed effects only under a very specific set of assumptions. However, in the robustness section we present the results of a Poisson model estimated with province-industry fixed effects.

provides information on the patent applications of over 80 countries, including title and abstract, priority, patent family and PCT links, citation links and technology class of the patent. Our data come from the April 2011 release for all patent applications to the EPO in the years 2002-2008. Specifically, we extracted the number of firms' patent applications from Marin (2012) and Lotti and Marin (2013), which provides numbers of patent applications filed between 1977 and 2011 by Italian firms registered in the AIDA dataset sourced by Bureau van Dijk. Marin (2012) applies a very accurate matching procedure to PATSTAT and AIDA datasets, enabling matching of more than 80 per cent of the patent applications submitted by Italian companies to the EPO, during the observation period.¹¹

We assign our patents to specific sectors and provinces (NUTS-3 regions) according to the firm sector and headquarters' location available from AIDA. We use the patent application date as the patent reference year. For robustness purposes we also estimated the model using the patent priority date obtaining very similar results (not shown but available under request). In order to reduce as many zero observations as possible, we aggregate our two-digit industries according to the OECD classification of sectors, i.e. into low tech (LT), medium-low tech (MLT), medium-high tech (MHT) and high tech (HT) (see Appendix 1, Table A1 for details). In addition, we consider KIBS, which correspond to computer and related activities, R&D, and other business services including engineering, architectural, legal and management consulting activities (Miles *et al.*, 1995). Our final variable is the annual number of patent (*PAT*) applications from firms located in each of the 103 Italian provinces, belonging to the four OECD sectors and the KIBS industry.

Data on R&D are extracted from the OECD Analytical Business Enterprise Research and Development database (ANBERD), which provides annual information on business R&D expenditure broken down into 60 manufacturing and service sectors. We consider data for Italy and for the time period 2002-2008. ANBERD data are at the two-digit industry level. In order to make them compatible with our patent data, we converted them to province (NUTS 3 region) and OECD sector levels (see Appendix 2). The final R&D variable (*R&D*) is log-transformed.¹²

Data on FDI come from the *fDi Markets* database, which tracks cross-border greenfield investments for all sectors and all countries worldwide. Greenfield FDI are defined as entry or expansion in the host country of a foreign firm or a plant belonging to a foreign enterprise, which

¹¹ For more details see Marin (2012) and Lotti and Marin (2013).

¹² This computation of the R&D variable has some limitations. First, it does not reproduce the actual geographical distribution of R&D expenditure. Second, it assumes that the industry composition in each province mirrors the industry composition in the region. Notice that in the present study the variable is used only as a control.

does not take account of inward FDI from mergers between domestic and foreign firms, or acquisition of domestic firms from foreign enterprises.¹³

The data extracted on Italian IGFDI include detailed information on type and motivation of investment projects, business function, project localization, cluster, sector, capital investment and job creation,¹⁴ on a yearly basis over the period 2003-2008 (for more details see: http://www.fdimarkets.com). The FDI variable is aggregated at the NUTS 3 regional level, for each year *T*. Table 2 presents some descriptive statistics for our *PAT*, *R&D* and FDI variables.

Table 2. Summary statistics (pooled sample)

Variable	Obs.	Mean	Std. Dev.	Min	Max
PAT	3605	4.912	18.76	0	342
R&D	3605	13.29	1.474	6.927	18.710
DFDI	3090	0.361	0.480	0	1
JOBS_FDI	3090	1.769	2.518	0	8.600
JOBS_FDI_MAN	3605	0.571	1.532	0	7.329
JOBS_FDI_SER	3605	1.166	2.221	0	8.600
KINV FDI	3090	1.524	2.212	0	7.849
KINV_FDI_MAN	3605	0.476	1.323	0	7.849
KINV_FDI_SER	3605	0.996	1.928	0	7.552

Note: we added a unit constant inside the log function before the transformation in log.

Table 3 reports the distribution of IGFDI projects by year, function (i.e. by type of activity located in Italy according to the *fDi Markets* classification) and top five FDI destination cities. It is worth mentioning the strong concentration of FDI within few sectors (sales, marketing and support, business services, retail, and manufacturing) and the largest provinces (Milan, Rome, Turin).

For the remaining independent variables, information on province-level population density, province-industry employment are based on census data provided by the Italian Statistical Institute (ISTAT). Annual employment and value added are from the ASIA database (i.e. *Archivio Statistico delle Imprese Attive*).

¹³ Recent works using the same dataset are the following: Castellani *et al.* (2012), Crescenzi *et al.* (2014) and Castellani and Pieri (2015).

¹⁴ Since *fDI Markets* includes both actual and predicted investment projects, the number of jobs created each year may refer to the actual or the expected number of jobs. Data on FDI are updated annually by a team of experts and analysts matching media and press news, balance of payments statistics and company-level information

	nve nose pi	0 vinces
Year	Number of FDI	Percentages
2003	115	12.13
2004	131	13.82
2005	140	14.77
2006	148	15.61
2007	170	17.93
2008	219	23.10
2009	25	2.64
Total	948	100.0
Function		
Business services	142	14.98
Construction	81	8.54
Customer contact services	6	0.63
Design, development and testing	27	2.85
Education and training	9	0.95
Electricity	28	2.95
Extraction	7	0.74
Headquarters	12	1.27
ICT and internet infrastructures	7	0.74
Logistic, distribution and transportation	56	5.91
Maintenance and servicing	4	0.42
Manufacturing	132	13.92
Recycling	2	0.21
Research and development	27	2.85
Retail	162	17.09
Sales, marketing and support	244	25.74
Technical support centre	2	0.21
Top five host cities		
Milan	267	28.16
Rome	106	11.18
Turin	45	4.75
Florence	21	2.22
Bologna	19	2.00

Table 3. Inward greenfield FDI distribution by year, function and top five host provinces

Finally, to identify large metropolitan regions, we use the concept of 'larger urban zones' (LUZ). Classification of Italian cities into larger urban zones is based on Eurostat-Urban Audit III data. LUZ are approximated to the functional urban region extending beyond the city core. In Italy, the 32 selected LUZ refer to the national capital (Rome), the regional capitals, and large (min. 250,000 inhabitants) and medium-sized (50,000-250,000 inhabitants) local labour systems.

Table 4 summarizes all the variables utilized in the econometric analysis, while Table 5 shows the correlation matrix.

Table 4. Variable description

Name	Source	Description
PAT	PATSTAT, EPO	Count number of patent applications in province P, sector S, year T
R&D	ANBERD, OECD	Log industrial R&D expenditures, in province P, sector S, year T
DFDI	fDi Markets	Dummy inward greenfield FDI in province P, year T
JOBS_FDI	fDi Markets	Log number of jobs created by FDI in province P, year T
KINV_FDI	fDi Markets	Log capital investment for FDI in province P, year T
JOBS_FDI_MAN	fDi Markets	Log number of jobs created by FDI in province P, year T, manufacturing
		industry
JOBS_FDI_SER	fDi Markets	Log number of jobs created by FDI in province P, year T, service
		industry
KINV_FDI_MAN	fDi Markets	Log capital investment of FDI in province P, year T, manufacturing
		industry
KINV_FDI_SER	fDi Markets	Log capital investment of FDI in province P, year T, service industry
DEN	Istat	Log population per squared km of province P land area, year T
EMPL	ASIA	Log employment in province P, year T
LUZ1	Eurostat	Dummy =1 if a province P includes a local labour system defined as a
		larger urban zone by Urban Audit III project
LUZ2	Eurostat	LUZ1 excluding the provinces of Milan and Rome
SPEC_MAN	Istat	Log value added in manufacturing industry, province P, year T over total
		value added in province P, year T
BORDER_EFFECT		Dummy for boundary provinces belonging to different regions
PRO_IND	Istat	Log value added per capita in manufacturing, province P, year T
SOCIAL_CAPITAL	Istat	Log of social capital (percentage of employees in social cooperatives
		over total), province P, year T
CAPITAL_MKT	Istat	Financing risk (dacay rate of cash-financing), province P, year T

Note: we added a unit constant inside the log function before the transformation in log.

Table 5. Correlation matrix

1 abi	U 3. U		ation	matri	A.										
	[1]	[2]	[3]	[4]	[5]	[6]	[7]	[8]	[9]	[10]	[11]	[12]	[13]	[14]	[15]
[1]	1.00														
[2]	0.35	1.00													
[3]	0.21	0.25	1.00												
[4]	0.29	0.28	0.93	1.00											
[5]	0.26	0.19	0.54	0.57	1.00										
[6]	0.31	0.27	0.77	0.86	0.18	1.00									
[7]	0.28	0.25	0.91	0.96	0.56	0.83	1.00								
[8]	0.25	0.17	0.52	0.55	0.97	0.16	0.56	1.00							
[9]	0.31	0.25	0.76	0.83	0.16	0.96	0.85	0.14	1.00						
[10]	0.33	0.73	0.34	0.37	0.23	0.36	0.34	0.20	0.34	1.00					
[11]	0.31	0.35	0.32	0.40	0.25	0.41	0.38	0.23	0.39	0.44	1.00				
[12]	0.11	0.23	0.08	0.00	0.08	-0.05	-0.01	0.06	-0.05	0.32	-0.05	1.00			
[13]	0.14	0.13	0.19	0.18	0.09	0.17	0.19	0.09	0.18	0.22	0.05	0.18	1.00		
[14]	-0.11	-0.24	-0.12	-0.11	-0.10	-0.08	-0.09	-0.08	-0.07	-0.30	-0.22	-0.38	-0.07	1.00	
[15]	-0.08	-0.15	-0.09	-0.10	-0.01	-0.13	-0.09	-0.01	-0.12	-0.18	-0.13	-0.13	-0.11	0.13	1.00

[1] PAT; [2] R&D; [3] DFDI; [4] JOBS_FDI; [5] JOBS_FDI_MAN; [6] JOBS_FDI_SER; [7] KINV_FDI; [8] KINV_FDI_MAN; [9] KINV_FDI_SER; [10] EMPL; [11] DEN; [12] SPEC_MAN; [13] PROD_IND; [14] SOCIAL_CAPITAL; [15] CAPITAL_MKT.

4. Results

Tables 6 to 8 present the results of our baseline random effects panel negative binomial model, while a series robustness checks are presented in Tables 9 to 11.¹⁵ In order to make the size of the effects comparable across the variables, in the tables are reported the marginal effects.

Table 6, Column 1, shows that for the whole economy the number of patents is positively related to lagged province and sectoral level R&D intensity, province employment, local population density, the LUZ status and manufacturing specialization of the province, but not to the IGFDI dummy. In Columns 2 and 3 the sample is split according to the patenting sector i.e. manufacturing and KIBS. In Column 2, the coefficient of DFDI remains not statistically significant, while in Column 3 it is significant at 5 per cent level. Patenting in manufacturing is related more to past R&D and agglomeration forces, whereas patenting in KIBS is related to the size of the NUTS3 region and to the arrival of foreign capital.¹⁶

)	8			
ESTIMATION METHOD:	Panel Negative Binomial RE Model				
	Total sample	Manufacturing	KIBS		
R&D _{T-1}	0.065***	0.073**	0.002		
	[0.019]	[0.031]	[0.062]		
EMPL _{T-1}	0.756***	0.711***	1.269***		
	[0.060]	[0.075]	[0.184]		
DFDI _{T-1}	0.029	0.038	0.311**		
	[0.035]	[0.056]	[0.136]		
DEN _{T-1}	0.330***	0.313***	0.126		
	[0.069]	[0.089]	[0.165]		
LUZ	0.253**	0.116	0.078		
	[0.118]	[0.148]	[0.271]		
SPEC_MAN _{T-1}	0.921***	0.816***	0.596		
_	[0.187]	[0.250]	[0.404]		
Geographic dummy (NUTS 1)	Yes	Yes	Yes		
Industry dummy	Yes	No	No		
Time dummy	Yes	Yes	Yes		
N. Obs.	3,090	1,854	618		
N. industry/provinces	515	309	103		
Alpha test (p-values)	0.000	0.000	0.000		

Table 6. The relationship between FDI (dummy) and patenting

*** Significant at 1%; ** significant at 5%; * significant at 10%; clustered standard errors at the province-industry level in parentheses. Estimates include a constant term.

¹⁵ The choice between a Negative Binomial (NB) and a Zero-Inflated Negative Binomial (ZINB) specification is made on the base of a Vuong test. The test statistics is standard normally distributed, with positive values favouring the ZINB model, and negative values favouring the NB model. In our case the value of the test statistics is -2.90 (and statistically significant), indicating that the NB specification is better suited for our data.

¹⁶ We also split the DFDI variable into two dummies – for manufacturing and services IGFDI. These two dummies are never statistically significant in the estimates. For reasons of space, we do not report these estimates here.

Table 7 excludes the FDI dummy and includes FDI intensity measured in terms of (actual or predicted) jobs created.

ESTIMATION METHOD:		Panel Negativ	ve Binomial	RE Model	
	Total sample	Manufacturing	KIBS	Manufacturing	KIBS
R&D _{T-1}	0.064***	0.074**	0.005	0.075**	0.001
	[0.019]	[0.031]	[0.062]	[0.031]	[0.062]
EMPL _{T-1}	0.752***	0.709***	1.245***	0.707***	1.240***
	[0.060]	[0.076]	[0186]	[0.076]	[0.187]
JOBS_FDI _{T-1}	0.008	0.008	0.053**		
	[0.007]	[0.011]	[0.026]		
JOBS_FDI_MAN _{T-1}				-0.001	0.012
				[0.011]	[0.026]
JOBS_FDI_SER _{T-1}				0.010	0.051*
				[0.011]	[0.026]
DEN _{T-1}	0.328***	0.311***	0.131	0.312***	0.134
	[0.069]	[0.089]	[0.165]	[0.089]	[0.165]
LUZ	0.250**	0.113	0.073	0.108	0.065
	[0.118]	[0.148]	[0.271]	[0.148]	[0.271]
SPEC_MAN _{T-1}	0.935***	0.825***	0.657	0.838***	0.688*
	[0.188]	[0.251]	[0.405]	[0.251]	[0.407]
Geographic dummy (NUTS 1)	Yes	Yes	Yes	Yes	Yes
Industry dummy	Yes	No	No	No	No
Time dummy	Yes	Yes	Yes	Yes	Yes
N. Obs.	3,090	1,854	618	1,854	618
N. industry/provinces	515	309	103	309	103
Alpha test (p-value)	0.000	0.000	0.000	0.000	0.000

Table 7. The relationship between FDI intensity (jobs created) and patenting

*** significant at 1%; ** significant at 5%; * significant at 10%; standard errors in parentheses. Estimates include a constant term.

The results in Column 1 show that patents are not related to the intensity of IGFDI, while Columns 2 and 3 show that a positive relationship emerges only for the case of patents in the KIBS industry. In particular, a unit increase in JOBS_FDI is related to a 5 per cent increase in KIBS patenting. Columns 4 and 5 show that only job creation related to FDI in services is significantly related to patenting in the KIBS sector. Thus, the positive link between FDI and innovation occurs only within services.

The results are similar when looking at the financial intensity of FDI. According to our estimates in Table 8, the larger the amount of service IGFDI, the higher is the related number of patents in the province's KIBS industry. The results of our estimations confirm that patenting in

manufacturing remains related to urbanization economies, while patenting in KIBS is related more to external knowledge transfer through IGFDI.¹⁷

This result can be explained in part by the distribution of IGFDI in Italy. *fDi Markets* data show that almost 80 per cent of investment projects registered in 2003-09 are in service activities¹⁸ and only 15 per cent are in manufacturing. The main activities in the former group are retail, sales and marketing, customer care, logistics, and design and R&D. Therefore, we would argue that the positive relationship between IGFDI and regional patenting capability is probably driven by foreign MNEs' demand for customized knowledge intensive services (i.e. software development, new business solutions, design, forecasting).

ESTIMATION METHOD:		Panel Negative Binomial RE Model					
	Total sample	Manufacturing	KIBS	Manufacturing	KIBS		
R&D _{T-1}	0.067***	0.078**	0.005	0.079**	-0.001		
	[0.019]	[0.030]	[0.060]	[0.031]	[0.061]		
EMPL _{T-1}	0.801***	0.753***	1.246***	0.755***	1.242***		
	[0.058]	[0.072]	[0.179]	[0.072]	[0.180]		
KINV_FDI_ _{T-1}	0.007	0.006	0.062**				
	[0.008]	[0.013]	[0.029]				
KINV_FDI _MAN _{T-1}				0.0005	0.026		
				[0.015]	[0.032]		
KINV_FDI_SER _{T-1}				0.039	0.059**		
				[0.013]	[0.030]		
DEN	0.0004***	0.0004	0.0003	0.0004***	0.0003		
	[0.0001]	[0.0001]	[0.0002]	[0.0001]	[0.0002]		
LUZ _{T-1}	0.280**	0.142	0.078	0.143	0.075		
	[0.117]	[0.146]	[0.270]	[0.147]	[0.271]		
SPEC_MAN _{T-1}	0.905***	0.805***	0.731*	0.800***	0.757**		
	[0.190]	[0.254]	[0.410]	[0.255]	[0.412]		
Geographic dummy (NUTS 1)	Yes	Yes	Yes	Yes	Yes		
Industry dummy	Yes	No	No	No	No		
Time dummy	Yes	Yes	Yes	Yes	Yes		
N. Obs.	3,090	1,854	618	1,854	618		
N. industry/provinces	515	309	103	309	103		
Alpha test (p-value)	0.000	0.000	0.000	0.000	0.000		

Table 8. The relationship between FDI intensity (capital investment) and patenting

*** Significant at 1%; ** significant at 5%; * significant at 10%; standard errors in parentheses. Estimates include a constant term.

Instrumental variable strategy. In this section we test for the potential endogeneity of FDI using an instrumental variable strategy. Provinces with higher levels of innovative activity may attract more foreign investments, thus generating a problem of reverse causality and endogeneity of the FDI variable. To address this issue, we adopt an instrumental variable approach using measures

¹⁷ The magnitude of the effect is also very similar to that of jobs created by IGFDI: a one-unit increase in KINV_FDI in services is related to a 6 per cent increase in the number of patents in KIBS sectors.

¹⁸ It is useful to remember that KIBS represent only a small share of total service activities.

of criminal activity as instruments. As suggested by recent studies (see, e.g. Al-Sadig 2009; Daniele and Marani 2010) there is a clear negative relationship between criminality and inward FDI since these activities tend to discourage foreign investment at the local level by increasing the risks and costs of doing business. Danakol *et al.* (2013; p. 13) also underline that "the argument for corruption rests in the view that domestic players are better acclimatized to the institutional arrangements of the host economy than outsiders, and that perhaps the most significant indicator of institutional quality and the business environment from the perspective of foreign investors is the extent of local corruption". This argument is supported by the US Department of State (U.S. Department of State 2012; p. 12) which suggests that "political violence is not a threat to foreign investments in Italy, but corruption, especially associated with organized crime can be a major hindrance, particularly in the South".

At the same time we believe that criminality, even though not totally uncorrelated with innovation, can affect innovative activity not directly but mainly indirectly, by hampering economic activity. Therefore, we assume a conditional exclusion restriction. We argue that if we control for the main factors affecting the economic activity included in the model, such as geographical and sectoral fixed effects, together with the other control variables that measure the level of economic activity of the province (e.g.: R&D expenditure and employment), we can eventually consider criminality as an exogenous variable with respect to the innovation performance.

We test different measures of criminality, including some indicators of the presence of criminal organizations such as *Mafia* and *Camorra*. According to our analysis, the best instrument is a variable MAFIA which measures the population share of those municipalities where city councils were dissolved because of infiltration of Mafia over the total population of the province in which the municipality is located. This indicator is correlated with inward FDI, but not patents. Table 9 reports estimates of an exponential mean model in the KIBS sector, in which the lagged FDI province variables are instrumented by the lagged measure of MAFIA in the province. In the first stage regression, the estimated coefficient of the instrument is negative and statistically significant. This supports the theoretical validity of our instrumentation strategy. Note that the related F statistics, which ranges between 3.11 and 10.96, reveals that the selected instrument is more satisfactory when referred to the IGFDI dummy, whereas it is weaker when referred to the intensity of IGFDI.

Table 9 shows that sign and statistical significance of the coefficients of the instrumented inward FDI variables are consistent with previous estimates, confirming the robustness of our main results. It is worth reminding that our exercise represents an attempt to mitigate the problem of

endogeneity of our main explanatory variable, but we are aware that to fully deal with endogeneity stronger instruments would be desirable.

ESTIMATION METHOD:	Exponential mea	n model with endogenous	regressors			
	KIBS					
R&D _{T-1}	1.484	1.701	1.428			
	[1.232]	[1.435]	[1.128]			
EMPL _{T-1}	-6.171	-9.458	-6.049			
	[4.290]	[7.085]	[4.244]			
DFDI_instrumented _{T-1}	35.423**					
	[15.139]					
JOBS FDI instrumented _{T-1}		8.138*				
		[4.489]				
KINV FDI instrumented _{T-1}			7.121**			
			[3.144]			
DEN _{T-1}	-0.602	-1.356	-1.153			
	[1.059]	[1.315]	[0.926]			
LUZ	-2.738	-3.568	-2.742			
	[1.808]	[2.349]	[1.690]			
SPEC MAN _{T-1}	0.396	6.447**	5.683**			
—	[2.145]	[3.	[2.373]			
Geographic dummy (NUTS 1) ^(a)	Yes	Yes	Yes			
Time dummy	Yes	Yes	Yes			
N. Obs.	618	618	618			
N. industry/provinces	103	103	103			
R ² of first stage regression	0.262	0.374	0.326			
MAFIA _{T-1}	-0.854**	-3.758*	-4.253**			
	[0.258]	[2.132]	[1.857]			
F-statistics	10.96	3.11	5.25			
Endogeneity test:						
Chi ²	25.64	6.08	6.08			
(p-value)	(0.000)	(0.013)	(0.013)			

Table 9. The relationship between FDI intensity and patenting: IV estimates

*** Significant at 1%; ** significant at 5%; * significant at 10%; standard errors are in parentheses. Estimates include also a constant term. ^(a) We excluded the South dummy to avoid collinearity with the MAFIA variable.

5. Robustness checks

The estimates in the previous section provide evidence of a positive relationship between inward FDI and the patenting activity in advanced business service firms. In this section, we test for the robustness of our findings in different ways.

First, we estimate the baseline equations using different econometric approaches. Table 10 shows the results for the main specification estimated through a 'fixed effects' panel Poisson model which controls for province-industry fixed-effects and time fixed-effects.

ESTIMATION METHOD:	Panel FE Poisson Model					
	KIBS	KIBS	KIBS	KIBS	KIBS	
R&D _{T-1}	0.133	0.121	0.122	0.199	0.111	
	[0.081]	[0.081]	[0.083]	[0.081]	[0.083]	
EMPL _{T-1}	-0.105	-0.198	-0.228	-0.153	-0.142	
	[0.965]	[0.959]	[0.971]	[0.962]	[0.970]	
DFDI _{T-1}	0.299***					
	[2.63]					
JOBS_FDI _{T-1}		0.051**				
		[0.023]				
JOBS_FDI_MAN _{T-1}			0.007			
			[0.021]			
JOBS_FDI_SER _{T-1}			0.059**			
			[0.023]			
KINV_FDI _{T-1}			•••	0.060**		
				[0.025]		
KINV_FDI_MAN _{T-1}					0.021	
					[0.027]	
KINV_FDI_SER _{T-1}					0.068**	
					[0.025]	
SPEC_MAN _{T-1}	0.768	0.925	0.877	1.022	1.107	
	[1.425]	[1.432]	[1.432]	[1.436]	[1.445]	
Time dummy	Yes	Yes	Yes	Yes	Yes	
N. Obs.	618	618	618	618	618	
N. industry/provinces	103	103	103	103	103	
Alpha test (p-value)	0.000	0.000	0.000	0.000	0.000	

Table 10. The relationship between FDI intensity and patenting: fixed effects Poisson

*** significant at 1%; ** significant at 5%; * significant at 10%; standard errors in parentheses. Estimates include a constant term.

The estimates suggest that our previous results on the impact of FDI are robust to the inclusion of province-industry fixed effects.¹⁹ Only IGFDI in services are positively related to higher patenting in KIBS, with a marginal effect between 0.05 and 0.06.

Second, we normalize our dependent variable, subtracting from the number of patent applications in province P, sector S, year T, the national mean (or the national median) by sector and year. Using this 'new' dependent variable, we can estimate linear panel models. The results of these estimates (not reported here for reasons of space, but available upon request) confirm the existence of a positive relationship between inward FDI and the patenting activity KIBS firms in the host region.²⁰

Third, we try to control for spatial effects in the FDI-patenting (in KIBS) relationship by including in the baseline specifications of Table 7 a new set of variables: (i) 20 NUTS-2 region

¹⁹ In this case, while using the Poisson specification, we impose the equidispersion property.

²⁰ In other unreported estimates, we also extended the time lag of our IGFDI variables to two and three years. Even though we still obtained positive coefficients, because of higher standard errors the parameters were not more statistically significant.

dummies instead of the four NUTS-1 dummies in order to control for finer spatial effects; (ii) a new LUZ variable – LUZ2 – constructed excluding the provinces of Rome and Milan; (iii) a dummy variable for regional border effect which takes the value 1 if two provinces belonging to different NUTS2 regions share common border, and 0 otherwise.

ESTIMATION METHOD:	Panel Negative Binomial RE Model					
			KIBS			
R&D _{T-1}	-0.019	-0.015	-0.019	-0.014	-0.020	
	[0.065]	[0.065]	[0.065]	[0.065]	[0.065]	
EMPL _{T-1}	1.378***	1.356***	1.354***	1.370***	1.363***	
	[0.178]	[0.180]	[0.181]	[0.178]	[0.179]	
DFDI _{T-1}	0.325**					
	[0.138]					
JOBS_FDI _{T-1}		0.054**				
_		[0.027]				
JOBS_FDI_MAN _{T-1}			0.017			
			[0.026]			
JOBS_FDI_SER _{T-1}			0.047			
			[0.027]			
KINV_FDI _{T-1}				0.062**		
				[0.031]		
KINV_FDI_MAN _{T-1}					0.029	
					[0.033]	
KINV_FDI_SER _{T-1}					0.055*	
					[0.030]	
DEN _{T-1}	0.229	0.231	0.230	0.224	0.222	
	[0.210]	[0.210]	[0.210]	[0.209]	[0.210]	
SPEC_MAN _{T-1}	1.331**	1.435**	1.446**	1.459***	1.472**	
	[0.591]	[0.588]	[0.590]	[0.588]	[0.590]	
PROD_IND _{T-1}	-0.204	-0.184	-0.174	-0.194	-0.180	
	[0.345]	[0.344]	[0.344]	[0.343]	[0.345]	
SOCIAL_CAPITAL _{T-1}	0.171	0.168	0.162	0.156	0.147	
	[0.104]	[0.104]	[0.105]	[0.104]	[0.105]	
CAPITAL_MKT _{T-1}	-0.058	-0.052	-0.047	-0.054	-0.050	
	[0.055]	[0.055]	[0.054]	[0.054]	[0.053]	
LUZ2	-0.045	-0.043	-0.039	-0.038	-0.029	
	[0.240]	[0.240]	[0.241]	[0.239]	[0.239]	
BORDER_EFFECT	0.155	0.160	0.167	0.160	0.169	
	[0.292]	[0.291]	[0.291]	[0.290]	[0.290]	
Geographic dummy (NUTS_2)	Yes	Yes	Yes	Yes	Yes	
Time dummy	Yes	Yes	Yes	Yes	Yes	
N. Obs.	618	618	618	618	618	
N. industry/provinces	103	103	103	103	103	

Table 11. The relationship between FDI intensity and patenting: spatial and local checks

*** significant at 1%; ** significant at 5%; * significant at 10%; standard errors are in parentheses. Estimates include also a constant term.

In order to capture unobserved heterogeneity related to the characteristics of the local/territorial system, we also include in our specifications: (a) a variable for the level of labour

productivity in manufacturing (PROD_IND) in province *P*; and (b) two variables capturing, respectively, the average level of social capital (SOCIAL_CAPITAL) and capital market/financial development (CAPITAL_MKT) in the province (see Table 4 for a description of the variables).

Table 11 reports the results of these estimates. Again, we find no significant change with respect to the previous specifications.

6. Conclusions

This paper provides an empirical investigation of whether inward greenfield FDI are related to higher sectoral innovation activity in the host province. Combining different data sources, we estimated panel count models, where the yearly number of patents in province P and sector S is regressed against a series of lagged FDI variables measuring both the presence and intensity of FDI.

Our results show a positive relationship between FDI and local patenting capability only for the service industry. In particular, we find that larger inward FDI in service activities positively influences local patenting activity in KIBS, whereas we do not find a significant relationship between innovation and IGFDI in the manufacturing activities The results are robust to the inclusion of spatial controls, fixed effects and to some extent the control of endogeneity.

The scenario that emerges seems to be one of foreign multinationals locating new and large investment projects in service-related activities (such as stores, retail or showroom offices, and business-related activities), which is stimulating innovative activity in advanced business-service industries (such as computer-related firms, software industries, legal and management consulting industries and engineering services), likely related to the provision of new and customized knowledge-intensive services.

These results have two policy implications. First, we provide evidence that attracting FDI helps to promote local innovation capability. The introduction of new activities from abroad seems to generate the creation of new ideas in the territory, which suggests that policy targeted at promoting FDI inflows could actually improve the competitiveness of local territories. Second, we found that the FDI-innovation relationship is not universal, but depends on the sectors receiving the FDI. In the case of Italy, foreign investments are mainly concentrated in services, and these activities seem to stimulate patenting activity of knowledge-intensive firms. Policies aimed at attracting services FDI can be expected to promote innovation in related, knowledge-intensive industries. However, we found no evidence that patenting in manufacturing is affected by the presence of foreign multinationals. Moreover, innovation in manufacturing seems to depend on urbanization economies, confirming the findings of previous studies (e.g. Carlino *et al.* 2007).

Therefore, innovation policies should not be the same for all sectors of the economy, rather they should be tailored depending on the type of targeted activity.

Our analysis has also some limitations. First, although panel data and the use of external instruments reduce the problem of endogeneity, in some models our instruments are proved to be weak; we believe that further work might be done to deal with the endogeneity issue. Second, in this paper we do not explicitly address spatial correlation between local patenting and FDI location. Future research could assess the potential spatial decay of knowledge spillovers using spatial statistics and suitable econometric techniques. Finally, in our study we use one-year time-lag for the FDI explanatory variable, and therefore we are able to identify only the short-term effects of the foreign investment on provinces' innovations. Future works might extend the analysis on the medium- long-term effects with longer time lags and panel dataset. Despite these limitations, we believe our study constitutes a useful first attempt to understand the economic impact of FDI location on innovation at the sub-national level, that provides a helpful set of empirical and policy insights.

References

- Aghion P., Blundell R., Griffith R., Howitt P., Prantl S.: The effect of entry on incumbent innovation and productivity. Review of Economics and Statistics 91(1), 20-32 (2009)
- Aitken B., Harrison A.: Do domestic firms benefit from direct foreign investment? Evidence from Venezuela. American Economic Review 89(3), 605-618 (1999)
- Al-Sadig A.: The effects of corruption on FDI inflows. Cato Journal 29(2), 267-294 (2009)
- Baltagi B. H.: Econometric analysis of panel data. Fourth Edition, John Wiley & Sons Ltd, Chichester, West Sussex, (2005)
- Beaudry, C., Schiffauerova, A.: Who's right, Marshall or Jacobs? The localization versus urbanization debate. Research Policy 38(2), 318-337 (2009)
- Bertschek I.: Product and process innovation as a response to increasing imports and foreign direct investment. Journal of Industrial Economics 43(4), 341-357 (1995)
- Blomström M., and Kokko A.: Multinational corporations and spillovers. Journal of Economic Surveys 12(3), 247–277 (1998)
- Bode E., and Nunnenkamp P.: Does foreign direct investment promote regional development in developed countries? A Markov chain approach for US states. Review of World Economics 147, 351-383 (2011)
- Brambilla I., Hale G., Long C.: Foreign direct investment and the incentives to innovate and imitate. Scandinavian Journal of Economics 111(4), 835-861 (2009)
- Branstetter L.: Is foreign direct investment a channel of Knowledge Spillovers? Evidence from Japan's FDI in the United States. Journal of International Economics 68(2), 325-344 (2006)
- Bitzer J., and Görg H.: Foreign direct investment, competition and industry performance. The World economy 32(2), 221-233 (2009)
- Blind K. and Jungmittag A.: Foreign direct investment, imports and innovations in the Service Industry. Review of Industrial Organization 25, 205-227 (2004)
- Carlino, G.A., Chatterjee, S., Hunt, R.M.: Urban density and the rate of invention. Journal of Economics 61, 389-419 (2007)
- Castellani D., Meliciani V., Mirra L.: The determinants of inward foreign direct investment in business services across European regions, Quaderni del Dipartimento di Economia, Finanza e Statistica 104, Università di Perugia, Dipartimento Economia (2012)
- Castellani D., Pieri F.: Outward Investments and Productivity: Evidence from European Regions. Regional Studies, forthcoming (2015)
- Cipollina M., Giovannetti G., Pietrovito F., Pozzolo A.F.: FDI and growth: What cross-country data say. The World Economy 35(11), 1599-1629 (2012)

- Crescenzi R., Pietrobelli C. and Rabellotti R.: Innovation drivers, value chains and the geography of multinational corporations in Europe. Journal of Economic Geography 14, 1053-1086, (2014)
- Danakol S.H., Estrin S., Reynolds P., Weitzel U.: Foreign direct investment and domestic entrepreneurship: blessing or curse? IZA Discussion Paper n. 7796, IZA (2013)
- Daniele V. and Marani U.: Organized crime and foreign direct investment: the Italian case. CESifo Working Paper n. 2416, Munich (2010)
- Driffield N.: Regional policy and spillovers from FDI in the UK. Annals of Regional Science 38, 579-594 (2004)
- European Commission: Innovation union scoreboard. European Union (2013)
- Feldman M.P., Audretsch D.B.: Innovation in cities: science-based diversity, specialization and localized competition. European Economic Review 43, 409-429 (1999)
- Girma S. and Wakelin K.: Local productivity spillovers from foreign direct investment in the U.K. electronics industry. Regional Science and Urban Economics 37, 399-412 (2007)
- Greenstone M., Hornbeck R., Moretti E.: Identifying agglomeration spillovers: Evidence from winners and losers from large plants openings. Journal of Political Economy 118(3), 536-598 (2010)
- Griliches Z.: Issues in assessing the contribution of Research and Development to productivity growth. Bell Journal of Economics 10(1), 92-116 (1979)
- Guimaraes P.: The fixed effects negative binomial model revisited. Economics Letters 99, 63-66 (2008)
- Haddad M. and Harrison A.: Are there positive spillovers from direct foreign investment? Journal of Development Economics 42, 51-74 (1993)
- Hausman J. A., Hall B. H., Griliches Z.: Econometric models for count data with an application to the patent-R&D relationship. Econometrica 59, 731-753 (1984)
- Haskel J., Pereira S., Slaughter M.: Does inward foreign direct investment boost the productivity of domestic firms? Review of Economics and Statistics 89(3), 482-496 (2007)
- Keller W.: International trade, foreign direct investment, and technology spillovers. NBER Working Paper n. 15442 (2009)
- Keller W., Yeaple S.: Multinational enterprises, international trade, and productivity growth: Firm level evidence from United States. Review of Economics and Statistics 91(4), 821-831 (2009)
- Kiriyama N.: Trade and Innovation. OECD Trade Policy Working Papers, n. 135 (2012)
- Knudsen, B., Florida, R., Stolarick, K., Gates, G.: Density and creativity in U.S. regions. Annals of the Association of American Geographers 98(2), 461-78 (2008)

- Jaffe, A.B., Trajtenberg, M., Henderson, R.: Geographic localization of knowledge spillovers as evidenced by patent citations. Quarterly Journal of Economics 108(3), 577-98 (1993)
- Javorcik B.: Does foreign direct investment increase the productivity of domestic firms? In search of spillovers through backward linkages. American Economic Review 94(3), 605-627 (2004)
- Lotti F. and Marin G.: Matching of Patstat Applications to AIDA Firms: Discussion of the Methodology and Results. Bank of Italy, Occasional Papers, n. 166 (2013)
- Marin G.: Matching of PATSTAT applications to AIDA firms Data set updated at February 2012 (2012). Available at: http://www.giovamarin.altervista.org/joomla/data
- Miles I., Kastrinos N., Flanagan K., Bilderbeek R., den Hertog P., Huntink W., Bouman M.: Knowledge Intensive Business Services: their roles as users, carriers and sources of innovation. Manchester, PREST Report to DG13 SPRINT-EIMS (1995)
- Peri G., Urban D.: Catching-Up to foreign technology? Evidence on the Veblen-Gerschenkron effect of foreign investment. Regional Science and Urban Economics 36, 72-98 (2006),
- UNCTAD: Training manual on statistics for FDI and the operations of TNCs. Volume 1, Geneva (2009)
- Vahter P.: Does FDI spur knowledge sourcing and innovation by incumbent firms? Evidence from manufacturing industry in Estonia. The World Economy 1308-1326 (2011)
- U.S. Department of State: 2012 Investment Climate Statement Italy. Bureau of Economic and Business Affairs (2012)

Appendix 1

Table A1, Sector conversion, from NACE to OECD classification		
Code	ATECO2002 (NACE Rev 1.1)	OECD classification
15-16	Manufacture of food products, beverages and tobacco	Low tech (LT)
17-18	Manufacture of textile and textile products	
19	Manufacture of leather and leather products	
20	Manufacture of wood and wood products	
21-22	Manufacture of pulp, paper; publishing and printing	
36-37	Manufacturing n.e.c.	
23	Manufacture of coke, refined petroleum and nuclear fuel	Medium-low tech (MLT)
25	Manufacture of rubber and plastic products	
26	Manufacture of other non-metallic mineral products	
27-28	Manufacture of basic metals and metal products	
24	Manufacture of chemical products and man-made fibres	Medium-high tech (MHT)
29	Manufacture of machinery and equipment n.e.c.	
34-35	Manufacture of transport equipment	
30-33	Manufacture of optical and electrical equipment	High-tech (HT)
72	Computer and related activities	KIBS
73	Research and Development	
74	Other business activities	

Table A1. Sector conversion: from NACE to OECD classification

Appendix 2. Constructing the yearly R&D variable at the province-sector level

Here, we describe the computation of the R&D variable used in the econometric estimates. Original industry R&D expenditure comes from the ANBERD database provided by the OECD and is available at the two-digit industry level of aggregation. In order to transform the data to the province-(OECD) industry level, we adopted the following strategy.

First, we used census data on regional firm demographics from the ASIA archive which provides yearly information on number of firms in each province and in each two-digit industry. Using the criteria in Appendix 1, we first pooled the two-digit industries into the four OECD sectors plus KIBS and then calculated a series of weights equal to the share of firms located in province P, OECD sector S and year T, in the total number of Italian firms in sector S and year T.

Following the same rule, we extracted R&D data (at the 2-digit level) from the ANBERD database and aggregated them at the level of the four OECD sectors plus KIBS, for each year. We then multiplied these yearly R&D data, initially available at NUTS 2 regional level, by the province-sector-year weight previously calculated. This provides us with a yearly R&D expenditure series for province P and (OECD) sector S across the period 2002-2009.

Finally, in order to test the representativeness of our new R&D data, we aggregated yearly province-sector values at the NUTS 2 region level, to obtain, year by year, regional industry R&D expenditure. These values were then compared with the values in the OECD STructural ANalysis (STAN) database of business enterprise R&D expenditure (in \$ million PPP) in Italy. For each of the years across 2002-2009, we found an average 0.9 correlation between our NUTS 2 region-level R&D variable and the value provided by OECD-STAN. We can conclude that our province-level disaggregation is a good representation of regional industry R&D expenditure in Italy.

RECENTLY PUBLISHED "TEMI" (*)

- N. 981 *The academic and labor market returns of university professors*, by Michela Braga, Marco Paccagnella and Michele Pellizzari (October 2014).
- N. 982 Informational effects of monetary policy, by Giuseppe Ferrero, Marcello Miccoli and Sergio Santoro (October 2014).
- N. 983 Science and Technology Parks in Italy: main features and analysis of their effects on the firms hosted, by Danilo Liberati, Marco Marinucci and Giulia Martina Tanzi (October 2014).
- N. 984 *Natural expectations and home equity extraction*, by Roberto Pancrazi and Mario Pietrunti (October 2014).
- N. 985 *Dif-in-dif estimators of multiplicative treatment effects*, by Emanuele Ciani and Paul Fisher (October 2014).
- N. 986 An estimated DSGE model with search and matching frictions in the credit market, by Danilo Liberati (October 2014).
- N. 987 Large banks, loan rate markup and monetary policy, by Vincenzo Cuciniello and Federico M. Signoretti (October 2014).
- N. 988 The interest-rate sensitivity of the demand for sovereign debt. Evidence from OECD countries (1995-2011), by Giuseppe Grande, Sergio Masciantonio and Andrea Tiseno (October 2014).
- N. 989 *The determinants of household debt: a cross-country analysis*, by Massimo Coletta, Riccardo De Bonis and Stefano Piermattei (October 2014).
- N. 990 How much of bank credit risk is sovereign risk? Evidence from the Eurozone, by Junye Li and Gabriele Zinna (October 2014).
- N. 991 The scapegoat theory of exchange rates: the first tests, by Marcel Fratzscher, Dagfinn Rime, Lucio Sarno and Gabriele Zinna (October 2014).
- N. 992 Informed trading and stock market efficiency, by Taneli Mäkinen (October 2014).
- N. 993 Optimal monetary policy rules and house prices: the role of financial frictions, by Alessandro Notarpietro and Stefano Siviero (October 2014).
- N. 994 Trade liberalizations and domestic suppliers: evidence from Chile, by Andrea Linarello (November 2014).
- N. 995 Dynasties in professions: the role of rents, by Sauro Mocetti (November 2014).
- N. 996 *Current account "core-periphery dualism" in the EMU*, by Tatiana Cesaroni and Roberta De Santis (November 2014).
- N. 997 Macroeconomic effects of simultaneous implementation of reforms after the crisis, by Andrea Gerali, Alessandro Notarpietro and Massimiliano Pisani (November 2014).
- N. 998 Changing labour market opportunities for young people in Italy and the role of the family of origin, by Gabriella Berloffa, Francesca Modena and Paola Villa (January 2015).
- N. 999 Looking behind mortgage delinquencies, by Sauro Mocetti and Eliana Viviano (January 2015).
- N. 1000 Sectoral differences in managers' compensation: insights from a matching model, by Emanuela Ciapanna, Marco Taboga and Eliana Viviano (January 2015).
- N. 1001 How does foreign demand activate domestic value added? A comparison among the largest euro-area economies, by Rita Cappariello and Alberto Felettigh (January 2015).
- N. 1002 *Structural reforms and zero lower bound in a monetary union*, by Andrea Gerali, Alessandro Notarpietro and Massimiliano Pisani (January 2015).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, v. 10, 6, pp. 1318–1347, **TD No. 698 (December 2008).**
- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate*, International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis*, Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704 (March 2009).**
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010)**.
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, v. 86, 2, pp. 269-283, **TD No. 743 (February 2010).**
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, Asset-based measurement of poverty, In D. J. Besharov and K. A. Couch (eds), Counting the Poor: New Thinking About European Poverty Measures and Lessons for the United States, Oxford and New York: Oxford University Press, TD No. 755 (March 2010).
- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**
- P. GUERRIERI and F. VERGARA CAFFARELLI, Trade Openness and International Fragmentation of Production in the European Union: The New Divide?, Review of International Economics, v. 20, 3, pp. 535-551, TD No. 855 (February 2012).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- R. GIORDANO and P. TOMMASINO, *Public-sector efficiency and political culture*, FinanzArchiv, v. 69, 3, pp. 289-316, **TD No. 786 (January 2011).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, v. 37, 12, pp. 2796-2813, **TD No. 800 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, TD No. 815 (July 2011).
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- S. GOMES, P. JACQUINOT, M. MOHR and M. PISANI, Structural reforms and macroeconomic performance in the euro area countries: a model-based assessment, International Finance, v. 16, 1, pp. 23-44, TD No. 830 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, v. 11, 5, pp. 1032-1052, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, in G. Gomel, D. Marconi, I. Musu, B. Quintieri (eds), The Chinese Economy: Recent Trends and Policy Issues, Springer-Verlag, Berlin, TD No. 838 (November 2011).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**
- R. GAMBACORTA and M. IANNARIO, *Measuring job satisfaction with CUB models*, Labour, v. 27, 2, pp. 198-224, **TD No. 852 (February 2012).**
- G. ASCARI and T. ROPELE, Disinflation effects in a medium-scale new keynesian model: money supply rule versus interest rate rule, European Economic Review, v. 61, pp. 77-100, TD No. 867 (April 2012).
- E. BERETTA and S. DEL PRETE, Banking consolidation and bank-firm credit relationships: the role of geographical features and relationship characteristics, Review of Economics and Institutions, v. 4, 3, pp. 1-46, TD No. 901 (February 2013).
- M. ANDINI, G. DE BLASIO, G. DURANTON and W. STRANGE, Marshallian labor market pooling: evidence from Italy, Regional Science and Urban Economics, v. 43, 6, pp.1008-1022, TD No. 922 (July 2013).
- G. SBRANA and A. SILVESTRINI, Forecasting aggregate demand: analytical comparison of top-down and bottom-up approaches in a multivariate exponential smoothing framework, International Journal of Production Economics, v. 146, 1, pp. 185-98, TD No. 929 (September 2013).
- A. FILIPPIN, C. V, FIORIO and E. VIVIANO, *The effect of tax enforcement on tax morale*, European Journal of Political Economy, v. 32, pp. 320-331, **TD No. 937 (October 2013).**

- G. M. TOMAT, *Revisiting poverty and welfare dominance*, Economia pubblica, v. 44, 2, 125-149, **TD No. 651** (December 2007).
- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, v.46, 4, pp. 693-713, **TD No. 730 (October 2009).**
- G. MICUCCI and P. ROSSI, *Il ruolo delle tecnologie di prestito nella ristrutturazione dei debiti delle imprese in crisi*, in A. Zazzaro (a cura di), Le banche e il credito alle imprese durante la crisi, Bologna, Il Mulino, **TD No. 763 (June 2010).**
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di politica economica, v. 105, 1, pp. 301-321, **TD No. 787 (January 2011).**
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, v. 6, 4, pp. 100-134, TD No. 791 (February 2011).
- P. ANGELINI, S. NERI and F. PANETTA, *The interaction between capital requirements and monetary policy*, Journal of Money, Credit and Banking, v. 46, 6, pp. 1073-1112, **TD No. 801 (March 2011).**
- M. BRAGA, M. PACCAGNELLA and M. PELLIZZARI, *Evaluating students' evaluations of professors,* Economics of Education Review, v. 41, pp. 71-88, **TD No. 825 (October 2011).**
- M. FRANCESE and R. MARZIA, Is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics, v. 15, 2, pp. 117-132, TD No. 828 (October 2011).
- L. GAMBACORTA and P. E. MISTRULLI, *Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?*, Journal of Money, Credit and Banking, v. 46, 4, pp. 753-778, **TD No. 829 (October 2011).**
- M. PERICOLI, *Real term structure and inflation compensation in the euro area*, International Journal of Central Banking, v. 10, 1, pp. 1-42, **TD No. 841 (January 2012).**
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance, v. 21, 2, pp. 324-344, TD No. 844 (January 2012).
- V. DI GACINTO, M. GOMELLINI, G. MICUCCI and M. PAGNINI, *Mapping local productivity advantages in Italy: industrial districts, cities or both?*, Journal of Economic Geography, v. 14, pp. 365–394, TD No. 850 (January 2012).
- A. ACCETTURO, F. MANARESI, S. MOCETTI and E. OLIVIERI, Don't Stand so close to me: the urban impact of immigration, Regional Science and Urban Economics, v. 45, pp. 45-56, TD No. 866 (April 2012).
- M. PORQUEDDU and F. VENDITTI, Do food commodity prices have asymmetric effects on euro area inflation, Studies in Nonlinear Dynamics and Econometrics, v. 18, 4, pp. 419-443, TD No. 878 (September 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, v. 76, 3, pp. 389-410, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, v. 12, 2, pp. 432-464, TD No. 886 (October 2012).
- M. TABOGA, *What is a prime bank? A euribor-OIS spread perspective*, International Finance, v. 17, 1, pp. 51-75, **TD No. 895 (January 2013).**
- L. GAMBACORTA and F. M. SIGNORETTI, *Should monetary policy lean against the wind? An analysis based on a DSGE model with banking*, Journal of Economic Dynamics and Control, v. 43, pp. 146-74, **TD No. 921 (July 2013).**
- M. BARIGOZZI, CONTI A.M. and M. LUCIANI, Do euro area countries respond asymmetrically to the common monetary policy?, Oxford Bulletin of Economics and Statistics, v. 76, 5, pp. 693-714, TD No. 923 (July 2013).
- U. ALBERTAZZI and M. BOTTERO, *Foreign bank lending: evidence from the global financial crisis,* Journal of International Economics, v. 92, 1, pp. 22-35, **TD No. 926 (July 2013).**

- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, v.8, 3, pp. 301-334, **TD No. 936 (October 2013).**
- D. PIANESELLI and A. ZAGHINI, *The cost of firms' debt financing and the global financial crisis*, Finance Research Letters, v. 11, 2, pp. 74-83, **TD No. 950 (February 2014).**
- A. ZAGHINI, *Bank bonds: size, systemic relevance and the sovereign*, International Finance, v. 17, 2, pp. 161-183, **TD No. 966 (July 2014).**
- M. SILVIA, Does issuing equity help R&D activity? Evidence from unlisted Italian high-tech manufacturing firms, Economics of Innovation and New Technology, v. 23, 8, pp. 825-854, TD No. 978 (October 2014).
- G. BARONE and S. MOCETTI, *Natural disasters, growth and institutions: a tale of two earthquakes,* Journal of Urban Economics, v. 84, pp. 52-66, **TD No. 949 (January 2014).**

2015

- G. BULLIGAN, M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors*, International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- A. CIARLONE, *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- G. BARONE and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- P. ALESSANDRI and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- R. AABERGE and A. BRANDOLINI, *Multidimensional poverty and inequality*, in A. B. Atkinson and F. Bourguignon (eds.), Handbook of Income Distribution, Volume 2A, Amsterdam, Elsevier, TD No. 976 (October 2014).
- M. FRATZSCHER, D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**

FORTHCOMING

- M. BUGAMELLI, S. FABIANI and E. SETTE, *The age of the dragon: the effect of imports from China on firmlevel prices*, Journal of Money, Credit and Banking, **TD No. 737 (January 2010).**
- G. DE BLASIO, D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds*, Industrial and Corporate Change, **TD No. 792 (February 2011).**
- A. DI CESARE, A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, **TD No. 831 (October 2011).**
- D. FANTINO, A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, TD No. 884 (October 2012).
- M. MARCELLINO, M. PORQUEDDU and F. VENDITTI, Short-Term GDP Forecasting with a mixed frequency dynamic factor model with stochastic volatility, Journal of Business & Economic Statistics, **TD No. 896 (January 2013).**
- M. ANDINI and G. DE BLASIO, Local development that money cannot buy: Italy's Contratti di Programma, Journal of Economic Geography, **TD No. 915 (June 2013).**
- J. LI and G. ZINNA, On bank credit risk: sytemic or bank-specific? Evidence from the US and UK, Journal of Financial and Quantitative Analysis, **TD No. 951 (February 2015).**